Author's response to reviews

Title: Lack of Treatment-Related Mortality Definitions in Studies of Children, Adolescents and Young Adults with Lymphomas, Solid Tumors and Brain Tumors: a Systematic Review

Authors:

Thai H Tran (thaihoa.tran@sickkids.ca)
Michelle Lee (michelleyk.lee@sickkids.ca)
Sarah Alexander (sarah.alexander@sickkids.ca)
Paul Gibson (paul.Gibson@lhsc.on.ca)
Ute Bartels (ute.bartels@sickkids.ca)
Donna Johnson (djohnston@cheo.on.ca)
Carol Portwine (portwc@mcmaster.ca)
Marianna Silva (silvam@kgh.kari.net)
Jason D Pole (jpole@pogo.ca)
Lillian Sung (lillian.sung@sickkids.ca)

Version: 2
Date: 25 March 2014

Author's response to reviews: see over
Dear Editors,

Please accept a revision of the manuscript entitled “Lack of Treatment-Related Mortality Definitions in Studies of Children, Adolescents and Young Adults with Lymphomas, Solid Tumors and Brain Tumors: a Systematic Review” for consideration for publication in your journal, BMC Cancer.

According to the editorial request, line numbering has been included in the main text. A change to one of the author’s name has also been made to fix a spelling mistake on the first submission.

Thank you very much for your time and consideration. We look forward to further correspondence with your office.

Sincerely,

Lillian Sung, MD, PhD